Author's response to reviews

Title: Validation of the Kidney Disease Quality of Life-Short Form: a cross-sectional study of a dialysis-targeted health measure in Singapore

Authors:

   Veena D Joshi Dr. (joshiveena0@gmail.com)
   Nandakumar Mooppil Dr. (nandakumar.m@nkfs.org)
   Jeremy FY Lim Dr. (jeremy.lim.f.y@sgh.com.sg)

Version: 2 Date: 2 June 2010

Author's response to reviews:

There are no revisions